Author's response to reviews

Title: Low-intensity blue-enriched white light (750 lux) and standard bright light (10 000 lux) are equally effective in treating SAD. A randomized controlled study. Ybe Meesters1*, Vera Dekker1, Luc J.M. Schlangen2, Elske H. Bos3 and Martine J. Ruiter4 1. University Center for Psychiatry, University Medical Center Groningen, The Netherlands 2. Philips Lighting, Eindhoven, The Netherlands 3. Interdisciplinary Center for Psychiatric Epidemiology, University Medical Center Groningen, The Netherlands 4. University of Groningen, Department of Clinical and Developmental Psychology, Groningen, The Netherlands Ybe Meesters*, University Center for Psychiatry, University Medical Center Groningen, P.O.Box 30 001, 9700 RB Groningen, The Netherlands Email: y.meesters@psy.umcg.nl Vera Dekker, University Center for Psychiatry, University Medical Center Groningen, P.O.Box 30 001, 9700 RB Groningen, The Netherlands Email: veradekker@gmail.com Luc J.M. Schlangen, Elske H. Bos (e.h.bos01@med.umcg.nl) Martine J. Ruiter (m.j.ruiter@psy.umcg.nl)

Authors:

Ybe Meesters (y.meesters@psy.umcg.nl)
Vera Dekker (veradekker@gmail.com)
Luc J.M. Schlangen (luc.schlangen@philips.com)
Elske H. Bos (e.h.bos01@med.umcg.nl)
Martine J. Ruiter (m.j.ruiter@psy.umcg.nl)

Version: 5 Date: 21 January 2011

Author's response to reviews:

The title page includes now the full names, institutional addresses, and e-mail addresses for all authors. The corresponding author is also be indicated.